Author’s response to reviews

Title: Treatment Selection of Early Stage Non-Small Cell Lung Cancer: The Role of the Patient in Clinical Decision Making

Authors:

Sahar Mokhles (s.mokhles@erasusmc.nl)
Joost Nuyttens (j.nuyttens@erasusmc.nl)
Mark de Mol (MdeMol1@amphia.nl)
Joachim Aerts (j.aerts@erasusmc.nl)
Alex Maat (a.p.w.m.maat@erasusmc.nl)
Ozcan Birim (o.birim@erasusmc.nl)
Ad Bogers (a.j.j.c.bogers@erasusmc.nl)
Hanneke Takkenberg (j.j.m.takkenberg@erasusmc.nl)

Version: 3 Date: 24 Oct 2017

Author’s response to reviews:

Editorial comments 1, 2 and 3 have been changed in our manuscript. Please let the corresponding author know whether the contributions of authors stands with reference to the guidelines of BMC Cancer. This project was a team effort therefore the contribution of every author was essential to finish this manuscript successfully.

Cover letter has been deleted, as it is not needed at this stage.